Author's response to reviews

Title: The capacity of short-chain fructo-oligosaccharides to stimulate faecal bifidobacteria: a dose-response relationship study in healthy humans

Authors:
Yoram Bouhnik (yoram.bouhnik@lrb.ap-hop-paris.fr)
Laurent Raskine (laurent.raskine@lrb.ap-hop-paris.fr)
Guy Simoneau (guy.simoneau@lrb.ap-hop-paris.fr)
Damien Paineau (d.paineau@nutri-health-sa.com)
Francis Bornet (frj.bornet@nutri-health-sa.com)

Version: 5 Date: 2 January 2006

Author's response to reviews:

Dear Sir,

You will find enclosed a modified version of the manuscript reference MS: 1982402604740638 entitled "The capacity of short-chain fructo-oligosaccharides to stimulate faecal bifidobacteria: a dose-response relationship study in healthy humans".

We made all format changes needed for publication.
We hope that the manuscript will be now suitable for publication in your review.

Yours faithfully

Prof. Yoram Bouhnik